



Article Using UMAP for Partially Synthetic Healthcare Tabular Data Generation and Validation

Carla Lázaro ^{1,*} and Cecilio Angulo ^{1,2,*}

- ¹ Intelligent Data Science and Artificial Intelligence Research Center, Technical University of Catalonia, Nexus II Building, Jordi Girona 29, 08034 Barcelona, Spain
- ² Robotics and Industrial Informatics Institute (CSIC-UPC), Llorens i Artigas 4, 08028 Barcelona, Spain
- * Correspondence: carla.lazaro@upc.edu (C.L.); cecilio.angulo@upc.edu (C.A.)

Abstract: In healthcare, vast amounts of data are increasingly collected through sensors for smart health applications and patient monitoring or diagnosis. However, such medical data often comprise sensitive patient information, posing challenges regarding data privacy, and are resource-intensive to acquire for significant research purposes. In addition, the common case of lack of information due to technical issues, transcript errors, or differences between descriptors considered in different health centers leads to the need for data imputation and partial data generation techniques. This study introduces a novel methodology for partially synthetic tabular data generation, designed to reduce the reliance on sensor measurements and ensure secure data exchange. Using the UMAP (Uniform Manifold Approximation and Projection) visualization algorithm to transform the original, high-dimensional reference data set into a reduced-dimensional space, we generate and validate synthetic values for incomplete data sets. This approach mitigates the need for extensive sensor readings while addressing data privacy concerns by generating realistic synthetic samples. The proposed method is validated on prostate and breast cancer data sets, showing its effectiveness in completing and augmenting incomplete data sets using fully available references. Furthermore, our results demonstrate superior performance in comparison to state-of-the-art imputation techniques. This work makes a dual contribution by not only proposing an innovative method for synthetic data generation, but also studying and establishing a formal framework to understand and solve synthetic data generation and imputation problems in sensor-driven environments.



Citation: Lázaro, C.; Angulo, C. Using UMAP for Partially Synthetic Healthcare Tabular Data Generation and Validation. *Sensors* **2024**, *24*, 7843. https://doi.org/10.3390/s24237843

Academic Editor: Ioannis Yiannis Kompatsiaris

Received: 9 October 2024 Revised: 18 November 2024 Accepted: 5 December 2024 Published: 8 December 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** physiological sensor data; synthetic data generation; data imputation; smart health; privacy preservation

1. Introduction

Synthetic data generation has emerged as a powerful tool in several domains that struggle with sensitive or scarce tabular data [1–3]. In particular, in the health domain, large amounts of data are continuously measured by sensors embedded in medical devices and monitoring systems. However, the sensitive nature of healthcare data and stringent privacy regulations make data sharing and use difficult. Synthetic data can mitigate these challenges by replacing the original data, reducing the risk of private data leakage while still retaining the ability to analyze and train models effectively. Furthermore, synthetic data generation effectively addresses data scarcity, missing data, and a lack of predictors by expanding the data set through artificial means, offering greater volume for analysis and model training.

There are many research areas where data availability presents unique challenges due to stringent privacy concerns, especially in the health domain [4–7]. These problems stem from the sensitive nature of medical information, such as diagnoses and treatments, which is often included in health records. However, such data are crucial for technological advancements in healthcare, which increasingly rely on accurate and comprehensive data

sets. In many healthcare applications, sensor data play a key role in collecting continuous patient information. Therefore, generating synthetic data to augment or replace sensorbased data sets could significantly reduce the reliance on direct measurements, offering a privacy-preserving alternative while maintaining analytical value [8].

Over time, various solutions have been proposed to address privacy risks, with early efforts focusing on anonymization and pseudo-anonymization. These processes entail the removal or encryption of identifiers that link an individual to stored data [9]. While these techniques are beneficial in numerous scenarios, they still present residual privacy risks. Consequently, the generation of synthetic data has been proposed as an alternative or supplementary methodology for the protection of privacy.

As an evolving field, synthetic data generation lacks a unified definition, which has resulted in a variety of interpretations across different contexts [10]. Most definitions agree that synthetic data entail the creation of new data values that mirror the statistical characteristics of the original data set [3,11]. However, a recent working definition, proposed by the Royal Society and The Alan Turing Institute, defines synthetic data as "data that has been generated using a purpose-built mathematical model or algorithm, with the aim of solving a (set of) data science task(s)" [12]. This divergence in definitions is particularly relevant for the validation of synthetic data, where the assessment can be based either on statistical fidelity or the data's utility in addressing data science challenges.

In [12,13], synthetic data are classified into two principal categories: fully synthetic and partially synthetic. Fully synthetic data sets are entirely artificial and contain no original data. In contrast, partially synthetic data sets replace only certain attributes with synthetic values, resulting in a hybrid data set that combines real and synthetic data [14]. This approach is particularly advantageous for handling sensitive attributes, where synthetic versions can be employed to substitute sensitive values while maintaining the utility of the non-sensitive data.

In cases where the challenge is not related to sensitive data but rather missing values, the focus shifts to data imputation. Data imputation is a widely accepted approach for estimating and filling in missing values based on available data [15]. Given the close relationship between partially synthetic data generation and data imputation, both approaches modify specific attributes while preserving the structure and integrity of the original data set.

The generation of synthetic data in clinical settings primarily involves tabular, time series, and text-based [10,16,17]. This article focuses on the generation of tabular data in the context of healthcare, introducing a novel methodology that leverages data visualization tools to enhance the generation of synthetic data. To the best of our knowledge, no existing methods employ visualization tools to analyze real-world reference data structures for the generation of partially synthetic data.

The goal of this work is to establish a new research direction in partially synthetic data generation. We aim to provide a foundation for the use of dimensionality reduction tools for data synthesis purposes. Ultimately, this line of research aims to extend to visualization-based fully synthetic data and provide a baseline for comparing this approach with more established generative models, including VAEs, GANs, and diffusion models.

This article is structured as follows. Section 2 provides an overview of background information and related work for the generation and imputation of synthetic data, with a particular emphasis on health-related techniques. Section 3 introduces the data sets used for developing the methodology, followed by a comparison of different visualization tools. The section also presents a formal framework to describe the problem and the proposed methodology. Finally, the section outlines the various workflow strategies employed to ensure data privacy. The proposed methodology is tested and validated in Section 4 on different cancer public repositories. Finally, Section 5 reviews the obtained results and discusses their limitations, while Section 6 concludes the article and suggests potential directions for future research.

2. Related Work

In this section, the state of the art regarding the topics addressed in the article is reviewed. Specifically, we examine and compare the concepts and methods associated with fully and partially synthetic data, explore various data imputation techniques, discuss data visualization and dimensionality reduction tools, and finally consider approaches to privacy preservation.

2.1. Fully and Partially Synthetic Data

The first proposals for the generation of synthetic data are often credited to Refs. [18,19]. These foundational works introduced methodologies based on statistical frameworks and techniques designed to address the issue of missing data. Since these initial proposals, many techniques have been developed to generate synthetic data, with objectives ranging from preventing privacy risks at disclosure to facilitating easier data access for machine learning models. This can involve either partial synthesis, where only some original records are replaced, or full synthesis, where the entire data set is substituted with a synthetic data set of the same size and composition [20].

The distinction between fully and partially synthetic data leads to the differentiation between fully and partially synthetic methods. Fully synthetic methods aim to replicate the statistical properties and relationships inherent in the original data without retaining any direct links to real individuals. While this approach provides robust privacy protection, it may compromise the data fidelity.

Conversely, partially synthetic methods seek to combine real and artificial data [21]. These data sets are typically used when only certain attributes are sensitive, allowing non-sensitive parts to remain unchanged. This approach seeks to balance privacy and data utility. However, partially synthetic data present greater privacy challenges than fully synthetic data [22]. For instance, research in [23] has demonstrated that partially synthetic data are more susceptible to membership inference attacks compared to fully synthetic data, which exhibit greater resilience to such vulnerabilities.

Several techniques have emerged for generating (fully) synthetic data. Among the most prominent techniques are variational autoencoders (VAEs), generative adversarial networks (GANs) and diffusion models (DM), which represent the cutting-edge approaches to synthetic data generation today.

Autoencoder neural networks are composed of two primary components: an encoder and a decoder. The encoder transforms input data into a latent space representation, while the decoder attempts to reconstruct the input from this representation with minimal error. Variational autoencoders (VAEs), introduced in 2013 [24], leverage the autoencoder framework to generate synthetic data.

While VAEs have shown remarkable success in generating images, adapting them for tabular data generation presents unique challenges. To address these challenges, several modifications of the original VAE architecture have been proposed for tabular data generation. One notable approach is Tab-VAE [25], specifically designed to handle the intricacies of tabular data. Further advancements in VAE-based models for tabular data include VAE-GMM [26], which integrates a Bayesian Gaussian mixture model (BGM) within the VAE architecture.

Although VAEs have been explored for synthetic tabular data generation, much of the recent research has focused on generative adversarial networks, first introduced in [27]. GANs consist of two neural networks: the generator and the discriminator, both trained in a competitive setting. The generator is responsible for creating synthetic data that resemble real-world data, while the discriminator's role is to assess the authenticity of these data.

Although GANs initially found success in synthesizing image data, recent efforts have adapted them for tabular data generation. One notable example is CTGAN [28], which modifies the GAN architecture to handle the specific challenges of tabular data, such as the coexistence of continuous and categorical variables. Another innovative example is PATE-GAN [29], which combines the Private Aggregation of Teacher Ensembles (PATE) framework with GANs to enforce privacy preservation during the data generation process. A recent proposal based on the GAN's generator–discriminator structure is NextConvGeN [30]. In this model, the generator uses data neighborhoods as input and produces synthetic samples within the convex space of these neighborhoods. The discriminator then evaluates these generated samples by contrasting them against a randomly selected batch of data from the broader data space.

In the healthcare domain, GANs have been applied to generate synthetic patient data. MedGAN [31], for instance, is designed to generate high-dimensional discrete variables, such as binary and count features, by combining an autoencoder with a GAN. This method is especially useful for simulating patient records with mixed data types. CorrGAN [32] builds on the MedGAN architecture by adding a correlation preservation mechanism, ensuring that the generated data maintain the original data attribute correlations. Further improvements to the GAN-based approach for medical data generation were proposed by Camino et al. [33], who introduced the Wasserstein GAN (WGAN) [34] with gradient penalty to stabilize training and improve the quality of the generated data.

Given the wide variety in medical data types, several methods have been proposed for generating multimodal synthetic data. For instance, [35] introduces the Multimodal Conditional GAN (MMCGAN), which combines a conditional GAN for tabular data generation with a model for conditional 3D image synthesis at variable resolutions.

Diffusion models [36,37] are generative models that employ a stochastic process to iteratively transform noise into meaningful data samples. These models have gained significant attention due to their remarkable performance across various generative tasks, particularly in text-to-image synthesis. Unlike GANs, which rely on adversarial training, diffusion models use a progressive denoising process, making them more stable and yielding higher-quality results in many cases, especially for complex data generation.

The impressive success of diffusion models in image synthesis has sparked interest in applying them to the generation of tabular data. One of the most promising examples is TabDDPM [38], a recent diffusion-based model designed specifically for tabular data. Another application of diffusion models can be found in the healthcare sector, where MedDiff [39] was introduced as a generative model for electronic health records (EHRs).

A final noteworthy development is the introduction of a fair diffusion model by Yang et al. [40]. This model is designed to address biases present in sensitive attributes of training data, making it suitable for generating balanced data sets that mitigate issues of fairness.

On the other side, different methods exist for generating partially synthetic data. For instance, a synthesizer algorithm called Perturbed Gibbs Sampler (PeGS) is proposed in [41], which employs a Markov chain Monte Carlo (MCMC) method to generate partially synthetic data.

In addition to classical techniques, machine learning-based methods have also been developed for partially synthetic data. In Ref. [42], a bidirectional recurrent neural network (BiRNN) is proposed to generate partially synthetic biomedical signals. In addition, a long-short-term memory (LSTM) algorithm is utilized in [43] to produce partially synthetic annotated text, which is then used to train a machine learning model for automatic deidentification of protected health information (PHI).

2.2. Data Imputation

The practice of data imputation is a commonly employed approach for estimating and filling in missing values within a data set, based on the existing data available for analysis [15]. Over the decades, numerous techniques have been developed to address this problem. Currently, both conventional statistical methods and advanced machine learning-and deep learning-based approaches are employed to tackle data imputation challenges.

Conventional methods encompass mean and median imputation and regression imputation, among others. These methods are classified as single imputation approaches, meaning they do not account for the uncertainty inherent in predicting missing values. However, multiple imputation (MI), originally proposed in [44], has been widely adopted to handle missing data due to its ability to produce unbiased results [45]. Multiple imputation entails the substitution of missing values in a data set with probable values, imputes, resulting in an imputed data set. This process is repeated multiple times, with slight variations incorporated to the model in each iteration [46], thus producing multiple imputed data sets. Then, the imputed data sets are used for the original data purpose, thereby yielding an individual analysis for each imputed data set. These analyses are then combined to produce an overall analysis [47].

The evolution of models used for obtaining imputes has progressed since the inception of multiple imputation. In recent studies, such as [15,48], different models for data imputation have been compared. These include multivariate imputation by chained equations (MICE), random forest algorithms, denoising autoencoders, and generative adversarial imputation nets (GAINs).

Given that imputed values are also synthetically generated values, there exist significant overlaps between the research domains of data imputation and synthetic data generation. Specifically, the outcome of a data imputation process essentially results in a partially synthetic data set. However, the expression "data imputation" is typically applied in the context of addressing missing data, whereas "synthetic" or "partially synthetic data" are commonly employed either to mitigate concerns about sensitive data or a data augmentation procedure.

Recognizing these similarities and overlap, a pioneering method for generating fully synthetic tabular data sets using imputation techniques was introduced in [49]. This procedure involves an iterative process to remove distinct sets of values, followed by the application of an imputation method to substitute the removed values with new ones. Following multiple iterations, the replaced values from each iteration are combined, resulting in the creation of a comprehensive table containing entirely synthetic data.

2.3. Data Visualization and Dimensionality Reduction Tools

The field of data visualization encompasses a range of techniques and tools that facilitate the comprehension of the structural organization and distributional characteristics of complex data sets. The visualization of high-dimensional data presents a significant challenge. To address this issue, data visualization techniques have been developed to reduce data dimensionality from highly dimensional space to a 2D or 3D representation. The most commonly utilized tools are t-distributed stochastic neighbor embedding (t-SNE) [50], Uniform Manifold Approximation and Projection (UMAP) [51], and TriMap [52].

Several studies have explored the relative strengths of these methods for specific applications. For example, in [53] statistical tests were applied to compare t-SNE and UMAP for single-cell data visualization but found no clear advantage of one method over the other. Similarly, Ref. [54] evaluated t-SNE, UMAP, and TriMap, along with other dimensionality reduction techniques, and concluded that each method impacts data sets differently. However, these studies suggest that selecting an appropriate dimensionality reduction method can significantly enhance the outcome.

A significant challenge that is common to all three techniques is the inherent trade-off between the preservation of local and global structures in the data [55]. Although there is no universally accepted definition of local or global structures, the term *local structure preservation* is used to describe the process of maintaining the set of high-dimensional neighbors for each data point in the lower-dimensional space. This encompasses the maintenance of relative distances or rank information among neighboring points, which essentially determines which points are closer in proximity to one another. In contrast, global structure preservation prioritizes the maintenance of relative distances or rank information between more distant points, thereby ensuring the reflection of broader patterns in the reduced dimensional space. In simpler terms, local structure preservation considers the distances between clusters.

A significant limitation of these methods, including t-SNE, UMAP, and TriMap, is the lack of a clear and systematic approach to parameter adjustment for balancing local and global structure preservation. This indicates that users must meticulously select between prioritizing either local or global structures, as it is not possible to optimize for both simultaneously.

Another crucial factor affecting the efficacy of these methodologies is the initialization strategy employed. t-SNE typically utilizes random initialization, UMAP applies a technique called Laplacian eigenmaps [56] and TriMap commences with principal component analysis (PCA). However, both UMAP and TriMap can be initialized randomly, as documented in the original papers [51,52]. Despite the tuning of hyperparameters, random initialization frequently results in suboptimal outcomes for both UMAP and TriMap [55,57].

Data visualization tools have demonstrated their utility in various health-related studies, particularly with UMAP. For instance, UMAP is used in [58] to visualize electronic health records (EHRs) from an emergency department, aiming to identify patient populations similar to a specific patient and provide additional information for the triage process. Similarly, UMAP is employed in [59] to extract patient features for detecting depression from data generated by wearable IoMT (Internet of Medical Things) devices. Furthermore, a UMAP-based methodology is proposed in [60] to identify minimal residual disease (MRD), a strong prognostic factor in two different types of leukemia.

UMAP has also shown significant capabilities in the efficient management and reduction of very high dimensional data. For example, in [61], UMAP is applied to five data sets with varying, high-dimensional structures: MNIST [62] (784 features), Fashion MNIST [63] (784 features), USPS [64] (256 features), Pen Digits [65] (64 features), and UMIST Face Cropped [66] (10304 features). Similarly, it is introduced in [67] a successful reduction from 6170 dimensions—representing genes in each of 1484 single-gene deletion strains—to a 2D space using UMAP. These studies showcase UMAP's adaptability in transforming highdimensional data into meaningful, low-dimensional representations without significant information loss.

On the other side, dimensionality reduction has also demonstrated significant value for imputation purposes. In [68], a novel technique was proposed to impute medical data records by applying a feature reduction method by class-based clustering. Another method, UMAP-SMOTENC [69] combines UMAP with the SMOTE-NC oversampling technique to work in a non-linear, compressed version of the data space. This approach effectively generates synthetic, privacy-preserving data while preserving essential patterns from the original data set, particularly those necessary for accurate classification tasks.

2.4. Privacy Preservation

Given the extensive use of healthcare data when developing algorithms for data synthetic generation and data imputation, several privacy threats have been identified in [70]: (i) identity disclosure or reidentification, where an adversary reveals the identity of a patient; (ii) attribute disclosure, where an adversary uncovers sensitive attributes of a patient; and (iii) membership disclosure, where an adversary successfully infers the presence of a patient in published data.

Ensuring privacy in the exchange of medical data among practitioners while protecting sensitive information has emerged as a critical challenge. Additionally, the monetary value of health data makes it a common target for attacks [22]. While synthetic data can mitigate some privacy issues by avoiding explicit one-to-one matches with real individuals, it can still inadvertently leak information related to training samples [23]. To address this risk, methods such as introducing statistical perturbation through differential privacy have been proposed [71].

3. Materials and Methods

In this section, the key components for developing the proposed methodology are presented. First, the data sets used in this study are introduced. This is followed by an

evaluation of the most appropriate visualization method for implementing the methodology. The section then outlines the proposed methodology, starting with the foundational concepts, followed by the detailed proposal for partially synthetic data generation. Finally, strategies for ensuring data privacy and workflow management between different data centers are discussed to address privacy concerns and facilitate secure data exchange.

3.1. Datasets

The proposed methodology is designed to be applicable to various topics within the health domain. In this case, the focus is specifically on prostate cancer and breast cancer data.

3.1.1. Prostate Cancer

In the case of prostate cancer, two public data sets are utilized: the Prostate Imaging and Cancer Analysis Information (PI-CAI) dataset, available at https://pi-cai.grandchallenge.org/DATA/ (accessed on 7 March 2024) and the Cancer Imaging Archive (CIA) dataset available at https://www.cancerimagingarchive.net/collection/prostate-mri-usbiopsy/ (accessed on 19 March 2024).

The PI-CAI data set comprises 1500 samples, each containing 12 features related to prostate cancer. The attribute case_csPCA is the primary outcome variable, indicating the presence of clinically significant prostate cancer. Certain features, such as patient_id, study_id, and mri_date, are excluded from this analysis due to their irrelevance for predicting case_csPCA and the potential privacy concerns they pose. Additionally, the attribute histopath_type, which refers to the procedure used to sample lesion tissue, is disregarded in this study because it does not directly impact the diagnosis of prostate cancer.

The attribute lesion_GS represents the Gleason Score (GS), a grading system that evaluates the aggressiveness of prostate cancer based on the microscopic appearance of cancer cells. Another important feature, lesion_ISUP, simplifies the Gleason Score into five distinct ISUP grades to better predict clinical outcomes. Furthermore, case_ISUP denotes the highest ISUP grade among the sampled lesions. Additional features include psa, which represents the prostate-specific antigen level, and psad, which is the prostatespecific antigen density. In cases where psad is missing, it can be computed by dividing PSA by the prostate volume (psad_computed).

Since case_ISUP reflects the highest risk grade in a sample, initially it is considered instead of lesion_ISUP. However, a correlation analysis revealed a substantial correlation coefficient (0.87) between case_ISUP and case_csPCA, leading to the exclusion of case_ISUP as a predictor. Consequently, patient_age, prostate_volume, and either psa, psad, or psad_computed are selected as predictors for the output case_csPCA.

The CIA data set, also a public repository, provides a wide range of information on prostate cancer. However, unlike the PI-CAI data set, the case_csPCA attribute is not available. Despite this limitation, the data set contains psa and prostate_volume values for each of its 24,783 samples. The attribute patient_age, however, is missing. Given the size and diversity of the CIA data set, it offers a valuable resource for exploring correlations and patterns in prostate cancer. The objective is to generate synthetic values for patient_age and assign synthetic case_csPCA values to each sample based on the available information.

3.1.2. Breast Cancer

This study utilizes two publicly available breast cancer datasets. The first dataset, referred to as BC-MLR, is the Breast Cancer dataset from the UCI (University of Chicago, Illinois) Machine Learning Repository, available at https://archive.ics.uci.edu/dataset/14/breast+cancer (accessed on 10 July 2024). The second dataset, BCNB, is the Early Breast Cancer Core-Needle Biopsy WSI dataset, accessible at https://paperswithcode.com/dataset/bcdalnmp (accessed on 8 July 2024).

The BC-MLR data set comprises 286 instances with nine attributes related to patient characteristics and tumor details. Patient-specific data include age at the time of diagnosis, represented by the age attribute, as well as the patient menopausal status at diagnosis,

noted in the menopause feature. Tumor-related information is captured through variables such as tumor size, which records the maximum diameter of the excised tumor, and deg-malig, reflecting the tumor degree of malignancy and aggressiveness.

Additional features in the BC-MLR data set address the extent of cancer spread, with inv-nodes indicating the number of lymph nodes involved in metastatic breast cancer, and node-caps providing information on whether the tumor has replaced the lymph and penetrated the capsule. Moreover, the data set records the affected breast side (breast) and the specific breast quadrant where the tumor is located (breast-quad). The data set also includes irradiat, which specifies whether the patient has undergone radiation therapy. The target variable in this data set is Class, which represents whether there is cancer recurrence after treatment.

On the other hand, the BCNB data set includes information from 1058 breast cancer patients. The age attribute records the patient age at the time of diagnosis, while tumor characteristics such as tumor size and tumor type provide detailed descriptions of the tumor. Additionally, the data set captures molecular markers such as the expression of estrogen receptor (ER), progesterone receptor (PR), and human epidermal growth factor receptor 2 (HER2), represented by the features ER, PR, and HER2, respectively. The data set also includes HER2 expression, which provides the results of an immunohistochemistry (IHC) test, used to predict how well the cancer will respond to treatment.

The BCNB data set further includes the histological grading feature, which assesses how closely the tumor cells resemble normal breast cells. Information regarding surgical interventions, specifically the type of surgery performed to remove nearby lymph nodes, is contained in the surgical feature. Other features of note include the Ki67 score, which measures the rate at which cancer cells are dividing, and the molecular subtype, a classification system grouping breast cancers based on shared characteristics and hormone receptor status. Finally, the data set provides information on lymph node involvement through features such as the number of lymph node metastases and ALN status, indicating whether the lymph nodes in the underarm area are affected.

Since ideally the database would have the same features, we focus on common features between the data sets. Therefore, particular attention is given to the age attribute, tumor size, and number of lymph node metastases, as these variables are present in both the BC-MLR and BCNB data sets.

3.2. Comparison of Visualization Tools for Database Representation

As discussed in Section 2.3, the most commonly utilized tools for 2D visualization of high-dimensional data include t-SNE, UMAP, and TriMap. Each of these methods has its advantages and limitations, particularly in their ability to preserve local and global structures.

In the context of this methodology, reproducibility and effective cluster visualization are of paramount importance. Specifically, we are interested in distinguishing clinically significant prostate cancer samples within the PI-CAI data set and cancer recurrence cases in the BC-MLR data set. Therefore, all three methods—t-SNE, UMAP, and TriMap—were applied to visualize the most comprehensive data set available for the two types of cancer, PI-CAI and BC-MLR, after normalizing the data to ensure consistency in the results.

Figure 1 presents the visualizations of the PI-CAI and BC-MLR data set using t-SNE, UMAP, and TriMap. Upon examination, it is evident that UMAP offers a more distinct separation of classes compared to the other methods. Neither t-SNE nor TriMap displays clear differentiation between the classes, highlighting UMAP's superior ability to organize the data meaningfully.

Based on these findings, this study focuses on UMAP as a dimensionality reduction tool. Aside from the observed capacity to effectively organize data in the form of clusters, it presents high reproducibility [72].



Figure 1. PI-CAI and BC-MLR visualizations using different dimensionality reduction techniques.

By default, UMAP embeds data into Euclidean space, typically in a 2D plane, which allows for clear and intuitive interpretations. Although UMAP allows for other embedding spaces, Euclidean space is specifically chosen for this study, as it facilitates the use of Euclidean distance for measuring similarities between data representations. Additionally, given the nature of medical data, periodic or other complex behaviors that would warrant the use of non-Euclidean spaces are not expected. The Euclidean space provides a more intuitive and appropriate framework for capturing the relationships inherent in medical data sets.

3.3. Proposed Methodology

The proposed procedure is designed to facilitate the exchange of medical data, with the primary goal of augmenting or completing data sets for use in data-driven solutions. Let us imagine hospital 1 with a private, complete, and small data set. On the other side there is hospital 2, with a larger data set, but in this case uncompleted. Our methodology allows hospital 2 to leverage the complete information in hospital 1 without sharing the actual data.

The methodology begins by extracting information from the complete data set in hospital 1, which is named the reference data, and transforming it into a two-dimensional space using UMAP. In a second step, using this transformed data map, along with the trained UMAP function, synthetic values are assigned to the incomplete data set (from hospital 2). These newly assigned synthetic values are then mapped back into the 2D representational space to evaluate their congruence with the reference data.

The proposed methodology is a general procedure able to generate artificial data which, as previously discussed, can be categorized into fully synthetic, partially synthetic, and imputed data. These different results are achieved trough different methods, as illustrated in Figure 2. There, it is highlighted how fully synthetic methods generate data sets that contain no real data from the reference or incomplete data sets. In contrast, partially synthetic methods produce data sets that combine real data from an incomplete data set with artificial data generated from the available information. Finally, imputation methods, which typically address missing data, result in data sets that contain a mix of real and artificial data. These imputation methods can either generate the missing values from only the actual incomplete data set (option A) or complement it with a reference data set containing real-world data (option B).



Figure 2. Representation of real (light gray) and artificial (dark gray) values for missing data (white) on fully synthetic, partially synthetic, and imputed data sets.

Since partially synthetic data aim to protect sensitive information and imputed data seeks to resolve missing data issues, the artificial data in the resulting data sets usually exhibit different distributions. Partially synthetic data typically include some synthetic features that are consistent across all entries. Conversely, the artificial values in an imputed data set do not necessarily correspond to the same feature for all entries.

The relevant elements involved in the definition of each type of problem and the steps in the proposed methodology are denoted as follows.

3.4. Formal Framework

In this subsection, we aim to provide a formal definition of the proposed framework for synthetic data generation. The following subsections introduce the real and synthetic datasets, formally define the problems addressed, establish their equivalence, and explore the role of visualization tools in the methodology.

3.4.1. Real-World Data Sets

Let $\mathcal{D}_{RW} = (\mathcal{X}, \mathcal{Y}) = \{(\mathbf{x}_i, \mathbf{y}_i)\}_{i=1}^N$ be a real-world data set, with N instances composed of d features $\mathbf{x} = (x^{(1)}, \dots, x^{(d)})$ and d_y outputs $\mathbf{y} = (y^{(1)}, \dots, y^{(d_y)})$. Values for some features can be unknown for all the N instances in the data set. Considering that known and unknown features are ordered, instances can be noted:

$$\mathbf{x}_i = (\mathbf{x}_i^K, \mathbf{x}_i^U) = \left(\left(x_i^{(1)}, \dots, x_i^{(d_K)} \right), \left(x_i^{(d_K+1)}, \dots, x_i^{(d)} \right) \right)$$
(1)

with \mathbf{x}_i^K and \mathbf{x}_i^U representing known (light gray cells in Figure 2) and unknown (white cells in the same figure) values for all the instances, respectively. It is important to note that unknown features may not necessarily be absent from the data set but are considered unknown due to privacy concerns, necessitating the generation of synthetic values for these sensitive features. Hence, the real-world data sets considered in the problem can be represented as

$$\mathcal{D}_{RW} = (\mathcal{X}^{K}, \mathcal{X}^{U}, \mathcal{Y}) = \{(\mathbf{x}_{i}^{K}, \mathbf{x}_{i}^{U}), \mathbf{y}_{i})\}_{i=1}^{N}$$
(2)

with dim(\mathcal{X}^K) = d^K , dim(\mathcal{X}^U) = d^U , and dim(\mathcal{Y}) = d_y . As discussed previously, two types of real-world data set are distinguished in this context: a reference data set and an incomplete data set. For the reference data set, \mathcal{D}_r , all values are known for all the features, represented by $d^K = d$. Conversely, in the incomplete data set, \mathcal{D}_i , values for at least one feature are missing, $d^U = d - d^K \ge 1$. Note that in both cases the output \mathcal{Y} can be known, unknown, or partially known.

3.4.2. Synthetic Data Sets

Let \mathcal{D}_g denote a synthetic data set (output data set), with $\hat{\mathbf{x}}_i^U$ denoting the synthetic values generated in some form for the unknown features of the *i*-th entry. Depending on the employed methodology for its generation, the output database can be fully synthetic,

$$\mathcal{D}_g = (\hat{\mathcal{X}}^U, \hat{\mathcal{Y}}) = \{ \hat{\mathbf{x}}_i^U, \hat{\mathbf{y}}_i \}_{i=1}^N$$
(3)

or partially synthetic,

$$\mathcal{D}_{g} = (\mathcal{X}^{K}, \hat{\mathcal{X}}^{U}, \mathcal{Y}) = \{(\mathbf{x}_{i}^{K}, \hat{\mathbf{x}}_{i}^{U}), \mathbf{y}_{i})\}_{i=1}^{N}$$

$$\tag{4}$$

Note that in the case of fully synthetic data, the output \mathcal{Y} must be necessarily generated $(\hat{\mathcal{Y}})$ because no incomplete data are used in the generation process. Conversely, for partially synthetic data, the output can either be real, denoted as $\mathcal{D}_g = (\mathcal{X}^K, \hat{\mathcal{X}}^U, \mathcal{Y})$, or synthetic, $\mathcal{D}_g = (\mathcal{X}^K, \hat{\mathcal{X}}^U, \hat{\mathcal{Y}})$, in cases where the output is unknown in the incomplete data and must therefore be generated. Both cases are considered later in the detailed description of the methodology.

3.4.3. Problem Definition

According to the provided notation, the three particular cases of the general synthetic data set generation problem can be defined as follows:

- 1. **Fully synthetic generation.** The proposed methodology aims to generate a fully synthetic data set, D_g , based on a real-world reference data set, D_r , with N_r instances and *d* features.
- 2. **Partially synthetic generation.** This approach also relies on a reference data set, D_r , with N_r instances and d features, along with an incomplete database with missing features. For simplicity, in this study we assume that dim $(\mathcal{Y}) = 1$ and only one feature (x) is missing in the incomplete data set, dim $(\mathcal{X}^U) = 1$, and this is also the only feature to be synthetically generated. The incomplete data set is denoted as $D_i = D_{r-\{x\}}$, which has N_i instances. Thus, there are d 1 known features and one unknown feature common to all instances. The goal is to generate a partially synthetic data set, D_g , which includes known features and synthetic values for one unknown feature.
- 3. **Imputation.** Imputation methodologies operate on an incomplete database with missing features, similar to the previous case. However, missing features may vary across entries. For instance, considering one missing feature for each entry, the incomplete data set is defined as $\mathcal{D}_i = \mathcal{D}_{r-\{x_j\}} = \{(\mathbf{x}_i^K, \mathbf{x}_i^U), \mathbf{y}_i\}_{i=1}^{N_i} - \{x_j\}, j \in \{1, 2, ..., d\}$. In this case, the imputation can rely on the non-missing values of the incomplete database (option A) or on a reference data set with no missing values (option B).

3.4.4. Problem Equivalence

A viable approach to generating fully synthetic data starts with the same initial setup as partially synthetic data generation: a reference data set D_r and an incomplete data set D_i . In this scenario, all features of the generated data should be synthetic. Therefore, the method for partially synthetic data generation (PSDG) can be iteratively applied, each time addressing a different missing feature until the generated data encompass all features. This approach is illustrated in Figure 3 (upper side), where each feature is synthesized in a separate iteration until a fully synthetic data set is obtained. In addition, imputation methods can be considered a specific case of partially synthetic data generation. Both approaches share a common initial structure with known and unknown features; however, the missing features in an imputation setup are not uniform across all entries. Consequently, the data set with missing values can be divided into multiple subsets, and imputation can be performed using partially synthetic methods, as depicted in Figure 3 (lower side).

Since both fully synthetic methods and imputation methods can be derived from a partially synthetic method, this paper focuses on developing a methodology for partially synthetic data. The proposed approach can be extended to the other two cases in a straightforward form.



Figure 3. Fully synthetic and imputed data generation from partially synthetic data generation (PSDG) methods. Real values (light gray), artificial values (dark gray), and missing data (white).

3.4.5. Visualization Tools

Finally, we propose the use of visualization tools as a valid methodology for PSDG. Therefore, some notation is introduced. Let $f_{umap,RW} : \mathcal{X} \longrightarrow \mathcal{C}_{RW}$ define a UMAP function trained with real-world data \mathcal{D}_{RW} , projecting the *d*-dimensional data input space \mathcal{X} to the latent space of 2D coordinates, \mathcal{C}_{RW} . Depending on the origin of the high-dimensional data, the resulting coordinates are denoted as $\mathcal{C}_r = f_{umap,r}(\mathcal{X})$, when the map corresponds to the reference data, or $\mathcal{C}_g = f_{umap,r}(\hat{\mathcal{X}})$ for the (fully, partially) synthetically generated data set. Note that for both cases, the UMAP function is trained on the reference data set, \mathcal{D}_r .

Let \mathcal{D}_v be the valid synthetic data, corresponding to the generated samples that are suitable and align well with the reference data, thus $\mathcal{D}_v \subset \mathcal{D}_g$. Specifically, the valid synthetic coordinates, $\mathcal{C}_v \subset \mathcal{C}_g$, are deemed valid when they share a common underlying structure with the reference coordinates, $\mathcal{C}_{r,i}$ in the 2D latent space:

$$\mathcal{C}_{v} = f_{umap,r}(\hat{\mathcal{X}}) \sim \mathcal{C}_{r} = f_{umap,r}(\mathcal{X})$$
(5)

3.5. Partially Synthetic Methodology Proposal

The proposed methodology aims to integrate both real-world reference data and incomplete data to generate a complete partially synthetic data set. This approach leverages a structured framework to ensure both data utility and privacy preservation. The overall workflow of the methodology is illustrated in Figure 4, providing a visual representation of the process steps and their interconnections.





Figure 4. Synthetic data generation framework proposal.

To operationalize this framework, we have developed an algorithm, presented as Algorithm 1. This algorithm outlines the detailed steps required to execute the methodology, including preprocessing, synthetic data generation, validation, and final data set compilation. In this section, the different steps presented in the algorithm are described.

As previously described, the initial setup consists of a reference database, D_r , and an incomplete database, which contains the same features as the reference data, except for the missing feature x, $D_i = D_{r-\{x\}}$.

Step 1: Preprocessing and setup

The first step, before applying the methodology, is to ensure uniformity in data structures between the incomplete and reference databases. This setup involves aligning units, sampling groups, data collection methodologies, and normalization techniques.

Let us assume that, from D_r , the range M of finite possible values for the feature x can be obtained. For categorical, logical, or ordinal variables, this process is straightforward. However, if the missing variable is continuous, a quantization preprocess is applied, which discretizes the continuous values into a finite set of intervals or levels.

Step 2: Synthetic data generation

Using the *M* possible values of the missing feature *x*, *M* different entries can be created for each sample out of N_i in \mathcal{D}_i by assigning all different values of the range to *x*. Consequently, a resulting generated database, $\mathcal{D}_g = \mathcal{D}_{r-\{x\}+\{\hat{x}\}}$, is obtained, where \hat{x} is the values assigned to the missing feature. Considering that the incomplete database, $\mathcal{D}_{r-\{x\}}$, contains N_i samples, \mathcal{D}_g contains $N_i \cdot M$ samples and *d* features (*d* – 1 of these are real-world features and one feature is synthetically generated).

Step 3: UMAP transformation

The core of the proposed methodology lies in utilizing the visualization tool UMAP to transform data from *d* dimensions to coordinates in two dimensions. The UMAP function is trained with \mathcal{D}_r , resulting in $f_{umap,r}$. This function enables the transformation of not only \mathcal{D}_r , but can also be applied to \mathcal{D}_g . By performing these transformations, the UMAP coordinates of the generated data, denoted as \mathcal{C}_g , can be obtained and compared to the coordinates of the reference data, \mathcal{C}_r .

Step 4: Reliability score assignment

Now, a validation step, defined as the reliability score assignment, determines which of the generated samples $\mathcal{D}_v \subset \mathcal{D}_g$ are suitable and which do not align well with the reference data. The comparison between samples involves calculating the mean value of the *x* feature for the *k*-nearest neighbors in C_r for the 2D coordinates in C_g . The mean value for the *k* neighbors around the *i*-th sample is denoted as $\overline{x}_{i,k}$. Validation entails comparing the \hat{x} value of instance $i(\hat{x}_i)$ in \mathcal{D}_g with $\overline{x}_{i,k}$. Depending on the feature disparity, defined

as the Euclidean distance $f_{\text{disp}} = ||\hat{x}_i - \overline{x}_{i,k}||$, the generated *i*-th sample obtains a certain reliability score, as outlined in Table 1.

Feature Disparity (f_{disp})	Reliability Score (r_c)
$f_{ ext{disp}} \leq \varepsilon$	1
$\varepsilon < f_{\rm disp} \le 2\varepsilon$	0.9
$2\varepsilon < f_{ m disp} \leq 3\varepsilon$	0.8
$3\varepsilon < f_{ m disp} \leq 4\varepsilon$	0.7
$4\varepsilon < f_{\rm disp} \le 5\varepsilon$	0.6
$5\varepsilon < f_{ m disp} \le 6\varepsilon$	0.5

Table 1. Reliability scores. Value ε depends on the range of the synthetic feature.

If the reliability score r_c of a certain sample exceeds a threshold, then the generated *i*-th sample is validated. For a single original sample in the incomplete database, $\mathcal{D}_{r-\{x\}}$, there may be multiple suitable values for *x*, resulting in the generation of additional samples. Therefore, the methodology not only imputes the real missing value, but also creates additional suitable samples, which constitute synthetic data.

Step 5: Cluster validation (if applicable)

For the last step, we can consider that the output is also unknown $\mathcal{D}_{r-\{x,y\}}$. Since the output is known for the reference data, instances can be grouped into clusters, so the output of the N_i entries can be assigned based on the distance between each generated coordinate \mathcal{C}_g and the cluster's centroid coordinates of \mathcal{C}_r . This process, known as cluster validation, involves defining a radius around each centroid until the density of the cluster samples within the radius falls below a predefined threshold. All samples from \mathcal{C}_g falling within these cluster areas are then assigned to the corresponding cluster $\mathcal{D}_g = \mathcal{D}_{r-\{x,y\}+\{\hat{x},\hat{y}\}}$. Conversely, samples lying outside any cluster area do not meet the criteria for cluster validation.

Adjusting parameters such as the sample density threshold for cluster validation, the number of neighbors considered for reliability score assignment (*k*), and the minimum reliability score for validated samples can significantly affect the outcome of the synthetic data generation process. These parameters should be carefully fine-tuned according to the specific objectives and requirements of the generated data.

Setting a minimum reliability score threshold for validated samples ensures that only high-quality synthetic data points are retained for downstream applications. This threshold serves as a filter to enhance the reliability and usefulness of the generated data.

Furthermore, the sample density threshold determines the extent of cluster validation by defining the area around each cluster centroid. By adjusting this threshold, one can control the inclusion of the cluster validation process. Similarly, varying the number of neighbors (k) considered for the reliability assessment of characteristics allows flexibility in evaluating the consistency of feature values among neighboring samples.

Step 6: Output

Upon completing the reliability score analysis and, if applicable, the cluster validation process, the algorithm produces a final set of synthetic samples. These samples are selected based on meeting the reliability score threshold, ensuring their consistency with the reference data. Additionally, for scenarios where the target variable is unknown, the algorithm assigns a predicted target value through the cluster validation process. The resulting data set consists of high-quality synthetic samples that are suitable for downstream applications.

Algorithm 1 Synthetic Data Generation with UMAP-based Dimensionality Reduction

Require: Complete reference dataset \mathcal{D}_r , incomplete dataset \mathcal{D}_i

Ensure: Validated synthetic dataset D_v

- 1: Step 1: Preprocessing and Setup
- 2: Align data structures: units, normalization, and sampling techniques.
- 3: Define range of possible values *M* for the missing feature *x* in \mathcal{D}_i .
- 4: **if** *x* is continuous **then**
- 5: Apply quantization to obtain discrete intervals for *M*.
- 6: end if
- 7: Step 2: Synthetic Data Generation
- Create a *generated dataset* D_g by assigning all *M* values to the missing feature *x* for each sample in D_i.
- 9: Size of \mathcal{D}_g : $|\mathcal{D}_g| = |\mathcal{D}_i| \times |M|$.
- 10: Step 3: UMAP Transformation
- 11: Train UMAP model $f_{\text{umap},r}$ on \mathcal{D}_r to obtain 2D coordinates \mathcal{C}_r .
- 12: Apply $f_{\text{umap},r}$ to \mathcal{D}_g to obtain 2D coordinates \mathcal{C}_g .
- 13: Step 4: Reliability Score Assignment
- 14: **for** each sample *i* in C_g **do**
- 15: Find *k* nearest neighbors in C_r .
- 16: Compute mean value $\overline{x}_{i,k}$ for the missing feature from *k* neighbors.
- 17: Calculate feature disparity $f_{\text{disp}} = ||\hat{x}_i \overline{x}_{i,k}||$.
- 18: Assign reliability score r_c based on f_{disp} using Table 1.
- 19: **if** $r_c \ge r_{\min}$ **then**
- 20: Validate sample *i* and add to \mathcal{D}_v .
- 21: end if
- 22: **end for**
- 23: Step 5: Cluster Validation (if applicable)
- 24: **if** output *y* is also missing **then**
- 25: Group reference data C_r into clusters.
- 26: **for** each sample j in C_g **do**
- 27: **if** *j* lies within cluster radius **then**
 - Assign cluster label and add to \mathcal{D}_v .
- 29: **end if**
- 30: **end for**
- 31: end if

28:

- 32: Step 6: Output
- 33: Return the validated synthetic dataset \mathcal{D}_v .

3.6. Data Privacy and Workflow Strategies

The proposed methodology is designed not only to augment or complete data sets, but also to ensure data privacy when the reference and incomplete data sets are located in different data centers. While the previous section detailed the methodology for general reference and incomplete data sets, this section focuses on the information workflow between separate data centers, emphasizing the protection of sensitive information.

To illustrate this, we revisit the scenario involving hospital 1 and hospital 2. In this case, hospital 2 (H_2) possesses a large but incomplete data set and seeks to leverage the complete data owned by hospital 1 (H_1) to fill in the missing information. Our methodology offers a solution to this challenge by enabling data exchange without compromising privacy, ensuring that sensitive data remain secure throughout the process.

Three scenarios are considered depending on the unknown information in the incomplete data set in H_2 :

• Case 0: The output is unknown, but the features are all known. In the case of missing outputs or classes, the validation process involves cluster validation, which requires the centroid cluster coordinates of D_r and a threshold radius from H_1 .

- Case 1: The output is known, but there is one missing feature. If the interest lies in assigning generated values for a missing feature, the validation process involves a reliability score assignment. This requires the mean value of the feature *x* for the *k*-nearest neighbors in C_r, for each instance *i* in C_g, *x*_{i,k}.
- Case 2: Both, the output and a feature are unknown. For scenarios involving the generation of both the unknown feature and the assignment of sample clusters, the validation process includes both steps.

Regardless of the scenario, all workflows require the UMAP function trained with the reference data in H_1 .

Our primary interest lies in case 1, which results in a synthetic feature for each entry. The other cases are presented for a better understanding of the system, but the workflow proposal focuses on case 1.

Depending on the interests and needs of the health institutions and their data centers, there exist different workflow options for sharing information, shown in Figure 5. The UMAP function, $f_{umap,r}$, trained with the data in H_1 , must be shared with hospital 2. This step does not pose any privacy risks because the reference data cannot be obtained back from the trained function.

The second information exchange occurs when hospital 2 transmits the UMAP coordinates, C_g , to the reference institution, hospital 1. Notably, in this exchange, H_1 does not obtain sensitive high-dimensional data due to the inherent limitations of UMAP coordinates. Since both the UMAP transformation, and its inverse, are stochastic processes, noise inevitably accompanies every transformation. Consequently, H_1 can only approximate the generated data from H_2 by inversely transforming the shared coordinates C_g . The information exchange in this step is crucial for computing the mean value of the feature x for the k neighbors in the reference coordinates for each generated sample.



Figure 5. Workflow options according to data privacy.

Finally, to perform the reliability score assignment, there are two options:

• **Option 1:** Hospital 2 shares the generated value \hat{x}_i for each sample with hospital 1. Hospital 1 calculates the feature disparity $f_{\text{disp}} = |\hat{x}_i - \overline{x}_{i,k}|$ and assigns a reliability score r_c to each generated sample. Hospital 1 then sends the reliability scores back to hospital 2 for them to keep the desired samples. • **Option 2:** Hospital 1 shares the mean neighbor value of feature *x* for each sample *i* with hospital 2. Hospital 2 calculates the feature disparity to provide the reliability scores.

4. Results

In this section, the proposed methodology is applied to generate synthetic data using the data sets introduced in Section 3.1. Since the true values for the missing data in the incomplete data sets are not available, direct validation of the generated synthetic data cannot be performed. As an alternative, a validation process is carried out using the PI-CAI data set. This involves partitioning PI-CAI into reference and incomplete data sets, allowing the generated synthetic values to be compared against the original values. This comparison facilitates a thorough evaluation of the performance and accuracy of the proposed methodology.

4.1. CIA Synthetic Data Generation from PI-CAI

In our experimental setup, we aim to generate synthetic samples for the CIA database using the PI-CAI data as a reference. Therefore, the reference data set comprises three features and a target variable per sample and the incomplete data set lacks one feature and the target variable.

Of particular importance is the consistency of features and units across databases. Notably, PSA (prostate-specific antigen) measurements in both the CIA and PI-CAI data sets are recorded in ng/mL, while prostate volume is documented in cc (cubic centimeters) for CIA and mL (milliliters) for PI-CAI, both equivalent units.

Before applying the UMAP transformation to the reference data, a data normalization is conducted to enhance visualization, specifically focusing on the age distribution in the UMAP plot, which is crucial for subsequent data validation. The chosen normalization method is Standard Scaler, as it yields the most favorable distribution post-UMAP transformation compared to other normalization techniques, especially regarding outliers.

After applying normalization and UMAP transformation to the reference data set, we ascertain the 2D coordinates of the PI-CAI reference samples. The resulting distribution has already been illustrated in Figure 1. As can be observed, aside from some outliers, the output variable facilitates the definition of clusters.

The patient_age feature in the reference database shows values from 35 to 92, therefore the number of possible values for the feature of interest is M = 58. By assigning to each sample on the incomplete database all the possibles ages, the resulting synthetic data set is composed of $58 \times 24,728 = 1,434,224$ potential entries.

The evaluation and validation of the generated data reveal several key findings. Cluster validation depends on the density threshold utilized to define the cluster area. Figure 6 illustrates the percentage of total generated samples that achieve cluster validation across various density thresholds, segmented by cluster and total count.

A marked reduction in the number of validated samples, particularly within the cluster case_csPCA= 0, is evident when the density threshold exceeds the value of 6. This significant decrease is also reflected in the total percentage of validated samples, which notably drops at this threshold value. This reduction indicates that for density thresholds greater than 6, the area around the cluster case_csPCA= 0 centroid fails to encompass the entire cluster. For the cluster case_csPCA= 1, this phenomenon occurs at a density threshold of 7.

Examining Figure 7, which depicts the cluster area for density thresholds set at 6 and 8, further elucidates the disparity in validated samples. It becomes evident that density values exceeding 6 result in a radius smaller than the cluster radius, consequently leaving some samples outside the cluster area. In such cases, synthetic samples situated at the periphery of the cluster would not pass the cluster validation process.



Figure 6. Percentage of cluster-validated samples for different density thresholds.



Figure 7. Cluster areas for density threshold equal to 10 (inner circle) and 8 (outer circle). Cluster centroid marked with a black \times .

Based on the aforementioned findings, the density threshold for this reference database is set to 6. Moving forward to the reliability score assignment, the outcomes depend on the selected number of neighbors (k) used to compute the mean age. It is also important to adjust the ϵ value for the concrete feature generation, in this case set to 1. Figure 8 illustrates the variation in the percentage of samples assigned specific reliability scores as k increases. Notably, there is a slight decrease in the percentage of samples as k increases. Conversely, as the reliability score decreases, there is a corresponding increase in the percentage of samples.



However, considering the most stringent reliability score ($r_c = 1$), the percentage of validated samples ranges between 20% and 30%, indicating that between 286,844 and 430,267 samples out of the 1,434,224 possible combinations are suitable.

Figure 8. Percentage of samples by reliability score for different number of considered neighbors.

4.2. BCNB Synthetic Data Generation from BC-MLR

To evaluate the reproducibility of the proposed methodology, the same procedure described in the previous section was applied to breast cancer data. In this experiment, the reference data set is the Breast Cancer Machine Learning Repository (BC-MLR) while the incomplete data set is the Breast Cancer Core-Needle Biopsy (BCNB) data set. As presented in Section 3.1, both data sets share three common features. Given the relatively small size of both data sets, the goal is not only to generate synthetic features, but also to augment the number of samples. In addition, since the BCNB data set is missing the output variable, the intention is also to obtain this target value.

Unlike in Section 4.1, where a specific variable was generated, the goal here is to generate several synthetic features for the BCNB data set. The only condition to generate a new feature and add it to the known features in BCNB is that they must be present in the BC-MLR. The correlation between the BC-MLR variables and the target features class is considered to determine which features to generate.

Table 2 shows the variables from BC-MLR and their corresponding correlations with the target attribute. The correlation between the known features is written in gray. Three other features with significant correlations with the target variable are chosen to be generated: node-caps, deg-malig, and irradiat.

The procedure followed is the same as described in Section 4.1, but in this case it is applied iteratively, generating one feature at a time. In the first iteration, the reference data set consists of the three known attributes, and it is projected into the UMAP space, as shown in Figure 9. In subsequent iterations, the attribute to be generated is added to the reference data set, resulting in a new UMAP representation for each iteration. Each time the reference data change, the density threshold to define the cluster area should be adjusted; in this case, for all iterations the density threshold was set to 1. This iterative approach allows the generation of multiple features, each producing a distinct UMAP projection based on the updated reference data.

Feature	Correlation			
age	-0.072			
menopause	0.052			
tumor-size	0.13			
inv-nodes	0.27			
node-caps	0.24			
deg-malig	0.3			
breast	-0.059			
breast-quad	0.037			
irradiat	0.19			

Table 2. Feature correlation with target variable. The correlation between the known features is written in gray. Significant feature correlations with the target variable are bolded.

The generation of different features also involves defining the value of ϵ for the reliability score assignment. In this case, $\epsilon = 1$ was adopted for the three features.



Figure 9. UMAP projection and cluster areas for the reference data in the first iteration. Cluster centroid marked with a black \times .

In the data generation process, the reliability score threshold is set to $r_c = 1$ for each iteration. The first feature to be generated is deg-malig, which can take a value of 1, 2, or 3, resulting in a total of $3 \times 1058 = 3174$ possible synthetic samples. Of these, 2558 samples were validated, accounting for 80.59% of the possible combinations.

The second generated feature is irradiat, which can take a binary value ('yes' or 'no'). Based on the previously generated data set containing 2558 samples, this step produces a data set of 5092 samples. Finally, the feature node-caps is generated, resulting in a final data set of 9427 samples.

By applying the proposed methodology iteratively, the initial incomplete data set with 1058 samples and three attributes is transformed into a partially synthetic data set with 8.9 times more samples (9427), six attributes, and the synthetic target feature (Class).

4.3. PI-CAI Validation

In order to validate the effectiveness of the algorithm, the ideal scenario would involve checking the actual attribute values of the incomplete data sets. For example, the patient_age values of the CIA database. However, this is not possible due to missing information. Therefore, the experimentation has been replicated for testing purposes by dividing the complete reference data set, PI-CAI, into two parts: PI-CAI 1 and PI-CAI 2.

To perform the validation, the complete and known PI-CAI database, with 1500 samples, was divided into five random groups, each containing 300 samples. One group was used as the reference data set (PI-CAI 1), while the other four groups served as incomplete data (PI-CAI 2). This process was repeated until all the data groups were utilized as reference data.

Using PI-CAI 1 as the reference data and PI-CAI 2 as the incomplete data, we applied our previously described methodology. By comparing the generated samples with the real data available in PI-CAI 2, we can evaluate the accuracy of the algorithm in imputing missing values and generating synthetic samples.

For each iteration, the methodology was applied, generating validated data. The validated data contained synthetic samples that did not correspond to any real patient and also correctly imputed data, by imputing the correct missing value of the incomplete data. After each iteration, the number of correctly imputed and synthetic generated data was recorded and averaged across the five iterations. This averaging process provides a more stable estimation of the performance of our algorithm.

In Figure 10, the amount of correctly imputed data is represented for each reliability threshold. As a reference, the number of samples originally corresponding to each age in the reference data set (PI-CAI 2) are represented in gray. Then, the number of correctly imputed samples per age are represented for the different reliability thresholds. For a more numerical interpretation, the percentage of correctly imputed data for certain age ranges is shown in Table 3. As expected, the number of correctly imputed samples increases as the reliability threshold decreases.

All the validated samples that do not correspond to real data are synthetic samples. In Figure 11, the number of generated samples per age depending on the reliability threshold can be observed. It is observed that the number of samples is augmented significantly. Also, as expected, ages with lower representation result in a lower number of generated samples.



Figure 10. Real samples from the incomplete database (light gray) and correctly imputed samples for different reliability values.

ge Range	35–44	45–54	55–64	65–74	75–84	85–92	Mean %
$r_{sc} = 0.5$	16.67	83.93	91.73	88.09	66.26	0.00	57.78
$r_{sc} = 0.6$	11.11	78.57	89.03	85.11	59.09	0.00	53.82
$r_{sc} = 0.7$	5.56	70.24	84.99	80.19	52.97	0.00	48.99
$r_{sc} = 0.8$	5.56	70.24	84.99	80.19	52.97	0.00	48.99
$r_{sc} = 0.9$	0.00	54.46	76.25	71.30	41.61	0.00	40.60
$r_{sc} = 1.0$	0.00	33.04	53.39	48.68	22.73	0.00	26.31
Mean	0	0	4.93	7.17	0	0	2.02
k-NN	0	0	1.30	0.69	0.70	0	0.45
	ge Range $r_{sc} = 0.5$ $r_{sc} = 0.6$ $r_{sc} = 0.7$ $r_{sc} = 0.8$ $r_{sc} = 0.9$ $r_{sc} = 1.0$ Mean k-NN	ge Range 35–44 $r_{sc} = 0.5$ 16.67 $r_{sc} = 0.6$ 11.11 $r_{sc} = 0.7$ 5.56 $r_{sc} = 0.8$ 5.56 $r_{sc} = 0.9$ 0.00 $r_{sc} = 1.0$ 0.00 Mean 0 k-NN 0	ge Range35-4445-54 $r_{sc} = 0.5$ 16.6783.93 $r_{sc} = 0.6$ 11.1178.57 $r_{sc} = 0.7$ 5.5670.24 $r_{sc} = 0.8$ 5.5670.24 $r_{sc} = 0.9$ 0.0054.46 $r_{sc} = 1.0$ 0.0033.04Mean00 k -NN00	ge Range35-4445-5455-64 $r_{sc} = 0.5$ 16.6783.9391.73 $r_{sc} = 0.6$ 11.1178.5789.03 $r_{sc} = 0.7$ 5.5670.2484.99 $r_{sc} = 0.8$ 5.5670.2484.99 $r_{sc} = 0.9$ 0.0054.4676.25 $r_{sc} = 1.0$ 0.0033.0453.39Mean004.93k-NN001.30	ge Range35-4445-5455-6465-74 $r_{sc} = 0.5$ 16.6783.9391.7388.09 $r_{sc} = 0.6$ 11.1178.5789.0385.11 $r_{sc} = 0.7$ 5.5670.2484.9980.19 $r_{sc} = 0.8$ 5.5670.2484.9980.19 $r_{sc} = 0.9$ 0.0054.4676.2571.30 $r_{sc} = 1.0$ 0.0033.0453.3948.68Mean004.937.17k-NN001.300.69	ge Range35-4445-5455-6465-7475-84 $r_{sc} = 0.5$ 16.6783.9391.7388.0966.26 $r_{sc} = 0.6$ 11.1178.5789.0385.1159.09 $r_{sc} = 0.7$ 5.5670.2484.9980.1952.97 $r_{sc} = 0.8$ 5.5670.2484.9980.1952.97 $r_{sc} = 0.9$ 0.0054.4676.2571.3041.61 $r_{sc} = 1.0$ 0.0033.0453.3948.6822.73Mean004.937.170 k -NN001.300.690.70	ge Range35-4445-5455-6465-7475-8485-92 $r_{sc} = 0.5$ 16.6783.9391.7388.0966.260.00 $r_{sc} = 0.6$ 11.1178.5789.0385.1159.090.00 $r_{sc} = 0.7$ 5.5670.2484.9980.1952.970.00 $r_{sc} = 0.8$ 5.5670.2484.9980.1952.970.00 $r_{sc} = 0.9$ 0.0054.4676.2571.3041.610.00 $r_{sc} = 1.0$ 0.0033.0453.3948.6822.730.00Mean004.937.1700 k -NN001.300.690.700

Table 3. Percentage of correctly imputed samples with the proposed methodology, mean and *k*-NN imputation.

For lower reliability values, a higher number of generated data is obtained. From the generated data, a percentage between 3.49% ($r_{sc} = 1$) and 2.79% ($r_{sc} = 0.5$) corresponds to correctly imputed data, the rest of the generated data are synthetic samples.



Figure 11. Number of synthetic samples per age.

Aside from the usability of the synthetic samples, the proposed methodology presents a higher imputation performance than the typical methods such as mean or *k*-NN imputation. As can be observed in Figure 12, the mean imputation, which assigns the mean value of the reference data for the missing variable to all the missing samples, only obtains a low percentage of correct imputation for the resulting mean ages. On the other hand, *k*-NN imputation calculates the mean values of the missing variable based on the nearest neighbors, identified by computing the Euclidean distance between the feature vectors of the samples. It can be observed that there are more ages with correctly imputed samples than with mean imputation, but the amount of correctly imputed samples is very low. Finally, by applying the proposed methodology with the most restrictive reliability score $r_{sc} = 1$, the number of correctly imputed samples increases significantly. A more detailed perspective can be obtained in Table 3 by comparing the exact percentage of correctly imputed samples.



Figure 12. Imputation performance of mean and *k*-NN (k = 10) vs. proposed methodology.

Given that traditional methods lack the capability to leverage complete data as reference information, a direct comparison with the proposed methodology is not entirely appropriate. However, for a general sense of performance, it is worth noting that even the most restrictive reliability score in our method results in a validated sample percentage above 25% while the traditional methods only reach 0.45% and 2.02%.

Also, a possible criterion to determine the reliability score is to consider the percentage of correctly imputed samples from the generated data. It can be seen that the percentage does not change between $r_c = 0.7$ and $r_c = 0.8$. However, the largest drop in the percentage occurs between $r_c = 0.9$ and $r_c = 1$.

5. Discussion

The proposed methodology has demonstrated a robust capability to generate synthetic data, achieving an augmentation of between thirteen and thirty times (depending on the selected reliability threshold) the amount of incomplete data in the context of prostate cancer databases. In the case of breast cancer, the original data sets are smaller, yet the methodology successfully yields a partially synthetic data set that is approximately nine times larger than the original. Notably, this augmentation includes the generation of three attributes along with the target variable.

Moreover, for the validation against the PI-CAI data set, the percentage of correctly imputed values underscores the effectiveness of the proposed methodology, particularly when compared to traditional imputation techniques such as mean imputation or k-nearest neighbors (*k*-NN) imputation.

However, the methodology relies on a reference data set, which poses a potential limitation. In practical use cases, especially in clinical settings, the availability of a well-structured reference data set may not always be guaranteed. If such a data set is absent, the performance of the methodology could be compromised. To address this issue, alternative approaches such as using publicly available databases or synthetic reference data sets could be explored.

One key advantage of generating this large volume of data is that it allows for machine learning models' generalization capabilities to be improved in low-data environments. The ability to generate high-quality synthetic data may reduce overfitting in models and allow for more robust decision making in clinical and other sensitive domains. Yet, the advantage and disadvantage of generating a large amount of data is that, without knowing the expected value, it becomes challenging to differentiate between imputed and synthetic samples. This ambiguity benefits synthetic data generation, as all the data can be utilized without privacy concerns, given that the small percentage of real data cannot be distinguished.

However, this methodology may not be suitable for simple imputation, as the correctly imputed samples are mixed with synthetic data. Despite this, since multiple values are generated for each entry, the method could be adapted for multiple imputation. Each generated value could be assigned to the missing data in different data sets, creating multiple imputed data sets for analysis. Given that the number of assigned values for each entry is not constant, adjustments should be made to the number of iterations for multiple imputation.

While this study applied UMAP for dimensionality reduction on a data set with relatively low dimensionality, reducing a maximum of six dimensions to two, the full potential of UMAP in handling very high dimensional data remains to be explored. UMAP is particularly powerful in applications involving hundreds of dimensions, such as genomic or proteomic data sets.

In summary, while the methodology offers clear advantages in terms of synthetic data generation and augmentation, its dependency on reference data and its potential limitations in simple imputation tasks warrant further investigation. Nonetheless, its broad applicability in fields with available reference data sets makes it a valuable tool for data augmentation, particularly in healthcare.

6. Conclusions

The aim of this research was to develop a methodology for generating data from an incomplete database by leveraging a complete database as reference data, while addressing privacy concerns. The proposed methodology successfully imputed missing data and generated synthetic samples, surpassing the number of incomplete data entries.

The methodology provides a secure framework for data augmentation by utilizing data from different centers without the need to transfer sensitive information. Additionally, it yields superior results for data imputation tasks.

The developed methodology has practical applications in generating partially synthetic data that do not correspond to any specific patient. Also, as discussed throughout the paper, a partially synthetic data generation method can be adapted to produce fully synthetic data sets and imputed data sets. Consequently, the proposed method can be applied to various data generation scenarios. This has significant implications for data-driven research and decision making in healthcare and other fields.

A secondary aim of this article was to provide a formal definition of the synthetic data generation problem, specifically addressing cases of partially synthetic data generation. This objective has been successfully achieved, as the paper articulates a clear and formal definition of the problem, thereby contributing to the ongoing discourse in the field of data synthesis.

In conclusion, the proposed methodology represents a valuable tool for generating synthetic data, providing a balance between data utility and privacy preservation. Further research can explore its application in other domains and investigate additional methods for enhancing its performance and scalability. In particular, testing alternative dimensionality reduction methods across diverse data sets with varying characteristics and higher dimensionalities would provide valuable insights into its adaptability and effectiveness. Additionally, by refining the methodology for fully synthetic data generation, its outcomes can be more directly compared to established methods, facilitating a comprehensive evaluation of its potential across fields.

Author Contributions: Conceptualization, C.L. and C.A.; methodology, C.L.; software, C.L.; validation, C.L. and C.A.; formal analysis, C.L. and C.A.; investigation, C.L. and C.A.; resources, C.L.; data curation, C.L.; writing—original draft preparation, C.L.; writing—review and editing, C.L. and C.A.; visualization, C.L.; supervision, C.A.; project administration, C.A.; funding acquisition, C.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the European Union Horizon Europe R&I program through the project FLUTE (Grant agreement no.: 101095382). It is also co-funded by the European Union–Next Generation EU funds, through the I+D+i PID2021-122835OBC21 research project ARTIFACTS (MCIN/AEI/10.13039/501000011033) and FEDER "Una manera de hacer Europa".

Institutional Review Board Statement: Not applicable.

Data Availability Statement: The original data presented in the study are openly available in the provided repository links.

Conflicts of Interest: The authors declare no conflicts of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

Abbreviations

The following abbreviations are used in this manuscript: VAEs Variational Autoencoders GANs Generative Adversarial Networks DM Diffusion Models BGM **Bayesian Gaussian Mixture** PATE Private Aggregation of Teacher Ensembles MCMC Markov Chain Monte Carlo LSTM Long-Short Term Memory MI Multiple Imputation MICE Multivariate Imputation by Chained Equations UMAP Uniform Manifold Approximation and Projection t-SNE t-distributed stochastic neighbor embedding PCA Principal Component Analysis EHR Electronic Health Record IoMT Internet of Medical Things PI-CAI Prostate Imaging and Cancer Analysis Information CIA Cancer Imaging Archive Gleason Score GS PSDG Partially Synthetic Data Generation PSA Prostate-Specific Antigen

References

- Vardhan, L.V.H.; Kok, S. Generating privacy-preserving synthetic tabular data using oblivious variational autoencoders. In Proceedings of the Workshop on Economics of Privacy and Data Labor at the 37th International Conference on Machine Learning, Virtual, 13–18 July 2020.
- Hernandez, M.; Epelde, G.; Alberdi, A.; Cilla, R.; Rankin, D. Synthetic data generation for tabular health records: A systematic review. *Neurocomputing* 2022, 493, 28–45. [CrossRef]
- 3. Hernandez, M.; Epelde, G.; Alberdi, A.; Cilla, R.; Rankin, D. Standardised metrics and methods for synthetic tabular data evaluation. *TechRxiv* 2021. [CrossRef]
- 4. Van Panhuis, W.G.; Paul, P.; Emerson, C.; Grefenstette, J.; Wilder, R.; Herbst, A.J.; Heymann, D.; Burke, D.S. A systematic review of barriers to data sharing in public health. *BMC Public Health* **2014**, *14*, 1144. [CrossRef] [PubMed]
- Dove, E.S.; Phillips, M. Privacy law, data sharing policies, and medical data: A comparative perspective. In *Medical Data Privacy Handbook*; Springer: Berlin/Heidelberg, Germany, 2015; pp. 639–678.
- Malin, B.; Goodman, K. Between access and privacy: Challenges in sharing health data. Yearb. Med. Inform. 2018, 27, 055–059. [CrossRef] [PubMed]
- Lange, L.; Wenzlitschke, N.; Rahm, E. Generating Synthetic Health Sensor Data for Privacy-Preserving Wearable Stress Detection. Sensors 2024, 24, 3052. [CrossRef]
- Kwon, G.S.; Choi, Y.S. Adjacent Image Augmentation and Its Framework for Self-Supervised Learning in Anomaly Detection. Sensors 2024, 24, 5616. [CrossRef]

- Vovk, O.; Piho, G.; Ross, P. Anonymization methods of structured health care data: A literature review. In Proceedings of the International Conference on Model and Data Engineering, Tallinn, Estonia, 21–23 June 2021; Springer: Berlin/Heidelberg, Germany, 2021; pp. 175–189.
- Giuffrè, M.; Shung, D.L. Harnessing the power of synthetic data in healthcare: Innovation, application, and privacy. NPJ Digit. Med. 2023, 6, 186. [CrossRef]
- 11. Raghunathan, T.E. Synthetic data. Annu. Rev. Stat. Its Appl. 2021, 8, 129–140. [CrossRef]
- 12. Jordon, J.; Szpruch, L.; Houssiau, F.; Bottarelli, M.; Cherubin, G.; Maple, C.; Cohen, S.N.; Weller, A. Synthetic Data—What, why and how? *arXiv* 2022, arXiv:2205.03257.
- 13. Surendra, H.; Mohan, H. A review of synthetic data generation methods for privacy preserving data publishing. *Int. J. Sci. Technol. Res.* 2017, *6*, 95–101.
- 14. Reiter, J.P. Inference for partially synthetic, public use microdata sets. Surv. Methodol. 2003, 29, 181–188.
- 15. Sun, Y.; Li, J.; Xu, Y.; Zhang, T.; Wang, X. Deep learning versus conventional methods for missing data imputation: A review and comparative study. *Expert Syst. Appl.* **2023**, 227, 120201. [CrossRef]
- Gonzales, A.; Guruswamy, G.; Smith, S.R. Synthetic data in health care: A narrative review. *PLoS Digit Health* 2023, 2, e0000082. [CrossRef] [PubMed]
- Kim, K.M.; Kwak, J.W. PVS-GEN: Systematic Approach for Universal Synthetic Data Generation Involving Parameterization, Verification, and Segmentation. Sensors 2024, 24, 266. [CrossRef] [PubMed]
- 18. Rubin, D.B. Statistical disclosure limitation. J. Off. Stat. 1993, 9, 461–468.
- 19. Little, R.J. Statistical analysis of masked data. J. Off. Stat. 1993, 9, 407-426.
- 20. Drechsler, J.; Haensch, A.C. 30 years of synthetic data. Stat. Sci. 2024, 39, 221–242. [CrossRef]
- 21. Murtaza, H.; Ahmed, M.; Khan, N.F.; Murtaza, G.; Zafar, S.; Bano, A. Synthetic data generation: State of the art in health care domain. *Comput. Sci. Rev.* 2023, *48*, 100546. [CrossRef]
- 22. Khan, S.; Hoque, A. Digital health data: A comprehensive review of privacy and security risks and some recommendations. *Comput. Sci. J. Mold.* **2016**, *71*, 273–292.
- Zhang, Z.; Yan, C.; Malin, B.A. Membership inference attacks against synthetic health data. J. Biomed. Inform. 2022, 125, 103977. [CrossRef]
- 24. Kingma, D.P. Auto-encoding variational bayes. arXiv 2013, arXiv:1312.6114.
- Tazwar, S.M.; Knobbout, M.; Quesada, E.H.; Popa, M. Tab-VAE: A Novel VAE for Generating Synthetic Tabular Data. In Proceedings of the ICPRAM, Rome, Italy, 24–26 February 2024; pp. 17–26.
- 26. Apellániz, P.A.; Parras, J.; Zazo, S. An improved tabular data generator with VAE-GMM integration. arXiv 2024, arXiv:2404.08434.
- Goodfellow, I.; Pouget-Abadie, J.; Mirza, M.; Xu, B.; Warde-Farley, D.; Ozair, S.; Courville, A.; Bengio, Y. Generative adversarial nets. *Adv. Neural Inf. Process. Syst.* 2014, 27.
- Xu, L.; Skoularidou, M.; Cuesta-Infante, A.; Veeramachaneni, K. Modeling tabular data using conditional gan. *Adv. Neural Inf. Process. Syst.* 2019, 32.
- Jordon, J.; Yoon, J.; Van Der Schaar, M. PATE-GAN: Generating synthetic data with differential privacy guarantees. In Proceedings
 of the International Conference on Learning Representations, Vancouver, BC, Canada, 30 April–3 May 2018.
- 30. Mahendra, M.; Umesh, C.; Bej, S.; Schultz, K.; Wolkenhauer, O. Convex space learning for tabular synthetic data generation. *arXiv* 2024, arXiv:2407.09789.
- Choi, E.; Biswal, S.; Malin, B.; Duke, J.; Stewart, W.F.; Sun, J. Generating multi-label discrete patient records using generative adversarial networks. In Proceedings of the Machine Learning for Healthcare Conference, PMLR, Boston, MA, USA, 18–19 August 2017; pp. 286–305.
- 32. Patel, S.; Kakadiya, A.; Mehta, M.; Derasari, R.; Patel, R.; Gandhi, R. Correlated discrete data generation using adversarial training. *arXiv* **2018**, arXiv:1804.00925.
- Camino, R.; Hammerschmidt, C.; State, R. Generating multi-categorical samples with generative adversarial networks. *arXiv* 2018, arXiv:1807.01202.
- Arjovsky, M.; Chintala, S.; Bottou, L. Wasserstein generative adversarial networks. In Proceedings of the International Conference on Machine Learning, PMLR, Sydney, Australia, 6–11 August 2017; pp. 214–223.
- Ziegler, J.D.; Subramaniam, S.; Azzarito, M.; Doyle, O.; Krusche, P.; Coroller, T. Multi-modal conditional gan: Data synthesis in the medical domain. In Proceedings of the NeurIPS 2022 Workshop on Synthetic Data for Empowering ML Research, New Orleans, LA, USA, 2 December 2022.
- Sohl-Dickstein, J.; Weiss, E.; Maheswaranathan, N.; Ganguli, S. Deep unsupervised learning using nonequilibrium thermodynamics. In Proceedings of the International Conference on Machine Learning, PMLR, Lille, France, 7–9 July 2015; pp. 2256–2265.
- 37. Ho, J.; Jain, A.; Abbeel, P. Denoising diffusion probabilistic models. *Adv. Neural Inf. Process. Syst.* 2020, 33, 6840–6851.
- Kotelnikov, A.; Baranchuk, D.; Rubachev, I.; Babenko, A. Tabddpm: Modelling tabular data with diffusion models. In Proceedings of the International Conference on Machine Learning, PMLR, Seattle, WA, USA, 30 November–1 December 2023; pp. 17564–17579.
- He, H.; Zhao, S.; Xi, Y.; Ho, J.C. MedDiff: Generating electronic health records using accelerated denoising diffusion model. *arXiv* 2023, arXiv:2302.04355.
- 40. Yang, Z.; Guo, P.; Zanna, K.; Sano, A. Balanced Mixed-Type Tabular Data Synthesis with Diffusion Models. *arXiv* 2024, arXiv:2404.08254.

- 41. Park, Y.; Ghosh, J. PeGS: Perturbed Gibbs Samplers that Generate Privacy-Compliant Synthetic Data. *Trans. Data Priv.* **2014**, 7, 253–282.
- 42. Hernandez-Matamoros, A.; Fujita, H.; Perez-Meana, H. A novel approach to create synthetic biomedical signals using BiRNN. *Inf. Sci.* **2020**, *541*, 218–241. [CrossRef]
- Libbi, C.A.; Trienes, J.; Trieschnigg, D.; Seifert, C. Generating synthetic training data for supervised de-identification of electronic health records. *Future Internet* 2021, 13, 136. [CrossRef]
- 44. Little, R.; Rubin, D. *Multiple Imputation for Nonresponse in Surveys*; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 1987; Volume 10, p. 9780470316696.
- 45. de Goeij, M.C.; Van Diepen, M.; Jager, K.J.; Tripepi, G.; Zoccali, C.; Dekker, F.W. Multiple imputation: Dealing with missing data. *Nephrol. Dial. Transplant.* **2013**, *28*, 2415–2420. [CrossRef]
- 46. Wayman, J.C. Multiple imputation for missing data: What is it and how can I use it. In Proceedings of the Annual Meeting of the American Educational Research Association, Chicago, IL, USA, 21–25 April 2003; AERA: Washington, DC, USA, 2003; Volume 2, p. 16.
- 47. Rubin, D.B. An overview of multiple imputation. In *Proceedings of the Survey Research Methods Section of the American Statistical Association;* ASA: Alexandria, VA, USA, 1988; Volume 79, p. 84.
- Getz, K.; Hubbard, R.A.; Linn, K.A. Performance of multiple imputation using modern machine learning methods in electronic health records data. *Epidemiology* 2023, 34, 206–215. [CrossRef]
- 49. Neves, D.T.; Alves, J.; Naik, M.G.; Proença, A.J.; Prasser, F. From missing data imputation to data generation. J. Comput. Sci. 2022, 61, 101640. [CrossRef]
- 50. Van der Maaten, L.; Hinton, G. Visualizing data using t-SNE. J. Mach. Learn. Res. 2008, 9, 2579–2605.
- 51. McInnes, L.; Healy, J.; Melville, J. Umap: Uniform manifold approximation and projection for dimension reduction. *arXiv* 2018, arXiv:1802.03426
- 52. Amid, E.; Warmuth, M.K. TriMap: Large-scale dimensionality reduction using triplets. arXiv 2019, arXiv:1910.00204.
- Roca, C.P.; Burton, O.T.; Neumann, J.; Tareen, S.; Whyte, C.E.; Gergelits, V.; Veiga, R.V.; Humblet-Baron, S.; Liston, A. A cross entropy test allows quantitative statistical comparison of t-SNE and UMAP representations. *Cell Rep. Methods* 2023, *3*, 100390. [CrossRef]
- 54. Zhao, X. Comparison of Data Visualization, Outlier Detection and Data Dimensionality Reduction Methods. *Highlights Sci. Eng. Technol.* **2024**, *85*, 1141–1149. [CrossRef]
- Wang, Y.; Huang, H.; Rudin, C.; Shaposhnik, Y. Understanding how dimension reduction tools work: An empirical approach to deciphering t-SNE, UMAP, TriMAP, and PaCMAP for data visualization. *J. Mach. Learn. Res.* 2021, 22, 1–73.
- 56. Belkin, M.; Niyogi, P. Laplacian eigenmaps and spectral techniques for embedding and clustering. *Adv. Neural Inf. Process. Syst.* **2001**, *14*.
- 57. Kobak, D.; Linderman, G.C. UMAP does not preserve global structure any better than t-SNE when using the same initialization. *bioRxiv* **2019**, 2019, 877522. [CrossRef]
- Hurley, N.C.; Haimovich, A.D.; Taylor, R.A.; Mortazavi, B.J. Visualization of emergency department clinical data for interpretable patient phenotyping. *Smart Health* 2022, 25, 100285. [CrossRef]
- Misgar, M.M.; Bhatia, M. Detection of depression from IoMT time series data using UMAP features. In Proceedings of the 2022 IEEE International Conference on Computing, Communication, and Intelligent Systems (ICCCIS), Kochi, India, 23–25 June 2022; pp. 623–628.
- 60. Weijler, L.; Kowarsch, F.; Wödlinger, M.; Reiter, M.; Maurer-Granofszky, M.; Schumich, A.; Dworzak, M.N. Umap based anomaly detection for minimal residual disease quantification within acute myeloid leukemia. *Cancers* **2022**, *14*, 898. [CrossRef]
- Allaoui, M.; Kherfi, M.L.; Cheriet, A. Considerably improving clustering algorithms using UMAP dimensionality reduction technique: A comparative study. In Proceedings of the International Conference on Image and Signal Processing, Mostaganem, Algeria, 24–25 November 2019; Springer: Berlin/Heidelberg, Germany, 2020; pp. 317–325.
- Deng, L. The mnist database of handwritten digit images for machine learning research [best of the web]. *IEEE Signal Process.* Mag. 2012, 29, 141–142. [CrossRef]
- 63. Xiao, H.; Rasul, K.; Vollgraf, R. Fashion-mnist: A novel image dataset for benchmarking machine learning algorithms. *arXiv* 2017, arXiv:1708.07747.
- 64. Hull, J.J. A database for handwritten text recognition research. IEEE Trans. Pattern Anal. Mach. Intell. 1994, 16, 550–554. [CrossRef]
- 65. Alpaydin, E.; Alimoglu, F. Pen-based recognition of handwritten digits data set. In *Machine Learning Repository*; University of California: Irvine, CA, USA, 1998; Volume 4.
- 66. Graham, D.B.; Allinson, N.M. Characterising virtual eigensignatures for general purpose face recognition. In *Face Recognition: From Theory to Applications*; Springer: Berlin/Heidelberg, Germany, 1998; pp. 446–456.
- 67. Dorrity, M.W.; Saunders, L.M.; Queitsch, C.; Fields, S.; Trapnell, C. Dimensionality reduction by UMAP to visualize physical and genetic interactions. *Nat. Commun.* **2020**, *11*, 1537. [CrossRef]
- 68. Yelipe, U.; Porika, S.; Golla, M. An efficient approach for imputation and classification of medical data values using class-based clustering of medical records. *Comput. Electr. Eng.* **2018**, *66*, 487–504. [CrossRef]
- 69. Almeida, G.; Bacao, F. UMAP-SMOTENC: A Simple, Efficient, and Consistent Alternative for Privacy-Aware Synthetic Data Generation. *Knowl. Based Syst.* 2024, 300, 112174. [CrossRef]

- 70. Chong, K.M. Privacy-preserving healthcare informatics: A review. In *Proceedings of the ITM Web of Conferences;* EDP Sciences: Les Ulis, France, 2021; Volume 36, p. 04005.
- Sablayrolles, A.; Douze, M.; Schmid, C.; Ollivier, Y.; Jégou, H. White-box vs. black-box: Bayes optimal strategies for membership inference. In Proceedings of the International Conference on Machine Learning, PMLR, Long Beach, CA, USA, 9–15 June 2019; pp. 5558–5567.
- 72. Becht, E.; McInnes, L.; Healy, J.; Dutertre, C.A.; Kwok, I.W.; Ng, L.G.; Ginhoux, F.; Newell, E.W. Dimensionality reduction for visualizing single-cell data using UMAP. *Nat. Biotechnol.* **2019**, *37*, 38–44. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.